Author's response to reviews

Title: Five year follow up of Japanese patients with Paget's disease of bone after treatment with a low dose of oral alendronate : a case series

Authors:

Kousuke Iba (iba@sapmed.ac.jp)
Junichi Takada (jtakada@kitago-ortho.com)
Takuro Wada (twada@sapmed.ac.jp)
Toshihiko Yamashita (tyamasit@sapmed.ac.jp)

Version: 4 Date: 12 January 2010

Author's response to reviews:

The JMCR Editorial Team

Dear Sir or Madam,

Thank you for the comments of the Deputy Editor. They were helpful in allowing us to revise our manuscript. We have put arrows in the two Xray figures and figure legends to show the pathological areas as the Editor indicated.

Please find enclosed the revised version of our manuscript entitled "Five-year follow-up of Japanese patients with Paget's disease of bone after treatment with a low dose of oral alendronate" (MS: 2037281320243354).

We would appreciate your kind consideration of its publication in Journal of Medical Case Report.

Sincerely Yours,

Kousuke Iba M.D., Ph.D.